Author’s response to reviews

Title: Evaluation of the chemical composition of nephrolithiasis using dual-energy CT in Southern Chinese gout patients

Authors:
Zhao-Xia Li (lisunglow_623@163.com)
Gen-Long Jiao (jiaogenlong@163.com)
Shu-Min Zhou (aileen426@qq.com)
ZhongYuan Cheng (825435580@qq.com)
Shoaib Bashir (859217811@qq.com)
Yi Zhou (zhou21yi@163.com)

Version: 2 Date: 24 Jun 2019

Author’s response to reviews:

Dear Zhang:

Thank you for your letter and reviewers’ comments concerning our manuscript entitled “Evaluation of the chemical composition of nephrolithiasis using dual-energy CT in Southern Chinese gout patients” (BNEP-D-19-00194R1). We have studied comments carefully and made corrections. Look forward for your good news.

Thanks again.

Yours

Yi Zhou